Author’s response to reviews

Title: Predictors of antibiotics co-prescription with antimalarial for patients presenting with fever in rural Tanzania

Authors:

Mustafa H Njozi (mnjozi@ihi.or.tz)
Mbaraka A Amuri MD (bamuri@ihi.or.tz)
Majige Selemani (mselemani@ihi.or.tz)
Irene M Masanja MD (imasanja@ihi.or.tz)
Brown Kigahe (bkigahe@ihi.or.tz)
Rashid Khatib PhD (rkhatib@ihi.or.tz)
Dan Kajungu (dan.kajungu@indepth-network.org)
Salim Abdula PhD (sabdula@ihi.or.tz)
Alexander N Dodoo PhD (alexooo@yahoo.com)

Version: 3 Date: 12 February 2013

Author’s response to reviews:

February 12, 2013
Dear editor,
Re: Covering letter for article resubmission
As the above heading suggests, I would like to resubmit my article to your journal concerning co-prescription of antimalarial with antibiotics in rural settings of Tanzania. I have made some changes as requested, changes include the ethical clearance reference number and how the consent was obtained. However changes have been done with the reference to abide with the journal format and the font type to Arial for better readability.
Attached with the article are tables and graphs with additional files of consent form used and questionnaires used for the study.
I think the article is in a better way now and follows journal instructions.
Thank you in advance
Mustafa Njozi